Author's response to reviews

Title: Pain and satisfaction during rigid cystoscopic ureteral stent removal: A preliminary study

Authors:

Jae Heon Kim (piacekjh@hanmail.net)
Sun Young Park (sunnypark97@schmc.ac.kr)
Moon Gyu Kim (rlaansrb1@schmc.ac.kr)
Hoon Choi (doc71377@naver.com)
Dan Song (backs7@schmc.ac.kr)
Seong Woo Cho (orsay2000@mac.com)
Yun Seob Song (yssong@schmc.ac.kr)

Version: 8  Date: 22 October 2014

Author's response to reviews:

Dear Editor

Now we have gained the trial registration number.

We feel very sorry about the delay of adding the trial registration number.

According to your suggestion, we added the trial number and as follows: Trial registration KCT0001260 (cris.nih.go.kr).

Thank you again for your patience.

Sincerely,

Jae Heon Kim